Author's response to reviews

Title: The M1 form of tumor-associated macrophages in non-small cell lung cancer is positively associated with survival time

Authors:

Junliang Ma (majunliangidiot@sina.com)
Lunxu Liu (lunxu_liu@yahoo.com.cn)
Guowei Che (chequowei@yahoo.com.cn)
Nanbin Yu (yunanbin@163.com)
Fuqiang Dai (dyfdtg@163.com)
Zongbing You (zyou@tulane.edu)

Version: 3 Date: 18 March 2010

Author's response to reviews: see over
March 17, 2010

Rachel Neilan, MSc
Assistant Scientific Editor
BMC-series Journals

Dear Editor Rachel Neilan,

Thank you for accepting our manuscript in principle.

**Reviewer:** Lill-Tove Busund  
**Reviewer’s report:**  
The manuscript has been improved. Reference 7 and 8 are identical.

**Response:** We have removed the previous reference #8, see page 4: citation of references in red text, and reference list #7 and 8. The rest of references are automatically corrected by EndNote.